Reviewer #1:
Scientific Quality: Grade B (Very good)
Language Quality: Grade B (Minor language polishing)
Conclusion: Accept (General priority)
Specific Comments to Authors: This manuscript is well organized and meets all of the requirements for publication. I recommend acceptance and publication of this manuscript.

1. Scientific Quality: Grade B (Very good)

Thanks for the comments, our study showed that, compared with BRAF wild type, BRAF\textsuperscript{V600E} mutation led to more angiogenesis and lymphoangiogenesis in the microenvironment. Meanwhile, the polarization of macrophages to M2 subtype was more obvious, and the immunosuppression was more prominent. Further analysis indicated that exosomes derived from BRAF\textsuperscript{V600E} mutant CRC cells were rich in certain lncRNAs and mRNAs, which might be linked to these alterations. This provides a hypothesis for finding new therapeutic strategies for BRAF\textsuperscript{V600E} mutant CRC.

2. Language Quality: Grade B (Minor language polishing)

Thanks for the comments, we have contacted the MedE Medical Editing and made some language changes again.